Author's response to reviews

Title: High cocoa polyphenol rich chocolate may reduce the burden of the symptoms in chronic fatigue syndrome

Authors:

Thozhukat Sathyapalan (thozhukat.sathyapalan@hyms.ac.uk)
Stephen Beckett (stephen.beckett@aol.com)
Alan S Rigby (asr1960@hotmail.com)
Duane Mellor (duane.mellor@hey.nhs.uk)
Stephen L Atkin (stephen.atkin@hyms.ac.uk)

Version: 2 Date: 30 July 2010

Author's response to reviews:

To 30/07/2010
The Editor
Nutrition Journal

Re: High cocoa polyphenol rich chocolate may reduce the burden of the symptoms in chronic fatigue syndrome.

Dear Editor,

Many thanks for asking us to revise this article and enclosing the constructive suggestions by the reviewers. We have now incorporated those suggestions that have markedly improved the manuscript. We hope that the revised manuscript is now suitable for publication in the Nutrition Journal. We enclose the detailed response to the reviewers comments.

Yours sincerely

T Sathyapalan